Author's response to reviews

Title: Soy foods have low glycemic and insulin response indices in normal weight subjects

Authors:

Robert M Blair Ph.D. (drblair@revivalsoy.com)
E. C. Henley Ph.D. (henleyec@aol.com)
Aaron Tabor M.D. (drtabor@revivalsoy.com)

Version: 4 Date: 20 December 2006

Author's response to reviews:

Dear Sirs,

As requested, I have revised the manuscript as outlined below:

1. Removed author qualifications
2. Included country in the affiliation details
3. Placed the website address in the reference list and
4. Used square brackets for citations in the text.

Additionally, we have made one other minor change. We have corrected the name of the soy chips used in the study from soy "pasta" chips to soy "protein" chips. This has been corrected throughout the manuscript and the figures where used.

The revised manuscript and figures have been uploaded and submitted.

Sincerely,

Bob Blair